

Delft University of Technology

COVID-19 Impact on the Oil and Gas Industry NO_2 Emissions A Case Study of the Permian Basin

A case cludy of the r entitlan basin

Serrano-Calvo, Raquel; Veefkind, J. Pepijn; Dix, Barbara; de Gouw, Joost; Levelt, Pieternel F.

DOI 10.1029/2023JD038566

Publication date 2023 Document Version Final published version

Published in Journal of Geophysical Research: Atmospheres

Citation (APA)

Serrano-Calvo, R., Veefkind, J. P., Dix, B., de Gouw, J., & Levelt, P. F. (2023). COVID-19 Impact on the Oil and Gas Industry NO Emissions: A Case Study of the Permian Basin. *Journal of Geophysical Research: Atmospheres, 128*(13), Article e2023JD038566. https://doi.org/10.1029/2023JD038566

Important note

To cite this publication, please use the final published version (if applicable). Please check the document version above.

Copyright

Other than for strictly personal use, it is not permitted to download, forward or distribute the text or part of it, without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license such as Creative Commons.

Takedown policy

Please contact us and provide details if you believe this document breaches copyrights. We will remove access to the work immediately and investigate your claim.



JGR Atmospheres

RESEARCH ARTICLE

10.1029/2023JD038566

Key Points:

- The NO₂ emissions calculated using TROPOMI data and the divergence method coincide with the downturn of O&G activity in the Permian basin
- Average NO₂ tropospheric column concentrations show a weaker decrease (-4%) in comparison to the significant reductions (~30%) observed in emissions during the COVID-19 lockdown
- We demonstrate a positive spatial and temporal relationship between oil and gas activity rates and emissions of NO₂ in the Permian basin

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

R. Serrano-Calvo, r.serranocalvo@tudelft.nl

Citation:

Serrano-Calvo, R., Veefkind, J. P., Dix, B., de Gouw, J., & Levelt, P. F. (2023). COVID-19 impact on the oil and gas industry NO₂ emissions: A case study of the Permian basin. *Journal of Geophysical Research: Atmospheres*, *128*, e2023JD038566. https://doi. org/10.1029/2023JD038566

Received 25 JAN 2023 Accepted 16 JUN 2023

© 2023. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

COVID-19 Impact on the Oil and Gas Industry NO₂ Emissions: A Case Study of the Permian Basin

Raquel Serrano-Calvo¹, J. Pepijn Veefkind^{1,2}, Barbara Dix³, Joost de Gouw^{3,4}, and Pieternel F. Levelt^{1,2,5}

¹Department of Geoscience and Remote Sensing, Civil Engineering and Geosciences, Technical University of Delft, Delft, The Netherlands, ²Royal Netherlands Meteorological Institute, GA De Bilt, The Netherlands, ³Cooperative Institute for Research in Environmental Sciences, University of Colorado, Boulder, CO, USA, ⁴Department of Chemistry, University of Colorado, Boulder, CO, USA, ⁵NCAR Atmospheric Chemistry Observations & Modeling Laboratory, Boulder, CO, USA

Abstract COVID-19 caused a historic collapse in fossil fuel demand, a general decline in economic activity, and hydrocarbon price volatility. This resulted in an unprecedented scenario to evaluate the contribution of the O&G (Oil and Gas) industry NO₂ (nitrogen dioxide) emissions in the Permian basin (United States), currently the second largest hydrocarbon-bearing area on Earth. TROPOMI (Tropospheric Monitoring Instrument), on board the Sentinel-5P satellite, has captured the impact of the oil and gas industry emissions during the COVID-19 lockdown. A generalized drop (~30%) of NO₂ emissions derived using the divergence method in comparison with 2019 was observed following the decline in production and drilling (13% and 68% respectively) during the lockdown. NO₂ tropospheric columns were less impacted with a smaller decrease (~4%) across the basins. This study demonstrates that the impact of the COVID-19 lockdown on NO₂ emissions was not only present in urban areas but also in vast O&G production regions, which shows the potential of TROPOMI to assess future pollution mitigation strategies for this industry.

Plain Language Summary The COVID-19 pandemic caused a big impact on the oil and gas industry, not only in production but also in price and demand. This situation was a good opportunity to analyze one of the most common gas emissions of this industry, nitrogen dioxide, in one of the biggest oil and gas production areas, the Permian basin. Using satellite imagery, it was observed a generalized drop in nitrogen dioxide emissions during the lockdown period of the pandemic. This study not only shows that the impact of the COVID-19 pandemic on nitrogen dioxide emissions was present in other environments apart from the urban for example, the oil and gas production regions, but also the capability to monitor it using satellite imagery.

1. Introduction

The COVID-19 lockdown has been an exceptional scenario to evaluate NO_2 emissions change due to the limitation of the human activity.

Since the appearance of the virus at the end of January 2020 in Wuhan (China), the COVID-19 impact on the energy industry was present in all aspects of this sector, from production to consumption. With the spread of the virus and the increase of cases around the world, strict lockdown policies were imposed on nearly 3 billion people globally (Rume & Islam, 2020), leading to an unprecedented reduction in fossil fuel demand and consumption. According to Le Quéré et al. (2020), aviation activity decreased by 75%, surface transport by 50%, and industry activity by 35%, producing a collapse in oil and gas consumption. Due to the decrease in consumption, the stockpile of gas and crude oil arrived at the limit in most of the productive countries which produced a decline in the barrel price to almost negative numbers. In response to this situation, OPEC (Organization of the Petroleum Exporting Countries) agreed to cut oil and gas production to recover the price of the barrel (Figure 1; Figure S1 in Supporting Information S1).

Linked to the decrease in fossil fuel consumption and COVID-19 lockdown policies, emissions of air pollutants and greenhouse gases were impacted worldwide (Archer et al., 2020; Barré et al., 2020; Ding et al., 2020; Gaubert et al., 2021; Keller et al., 2021; Tang et al., 2021; Turner et al., 2020). This included the emission of nitrogen dioxides ($NO_x = NO + NO_2$), which is emitted by the production, storage, and transportation of oil and gas. NO_x (Nitrogen oxides) is generated by combustion engines used to generate power, drilling, flaring, and for transportation of equipment and consumables. It is important to monitor and reduce NO_x emissions because, they affect





10.1029/2023JD038566



Figure 1. Time series of gas production, oil production and rig count in the Permian basin (U.S.) from January 2010 to August 2020 (Source of the data used: U.S. Energy Information Administration).

human health, vegetation, the chemistry balance in soils and waters and NO_2 is a precursor gas for tropospheric ozone and aerosols (Honour et al., 2009; Huang et al., 2021).

COVID-19 provides an unprecedented scenario to evaluate the NO_x emission of the oil and gas (O&G) industry for the largest oil and gas-producing region of the United States. In this study, we use satellite data collected by TROPOMI (TROpospheric Monitoring Instrument) (Veefkind et al., 2012) on board of Sentinel-5P and O&G activity data from the Permian basin region from January 2019 to October 2020. This research shows for the first time the impact of drastically reducing the oil and gas activity on concentrations and emissions of NO₂ in the Permian basin, not only in quantitative values but also in spatial and time relationships with the source.

2. Materials and Methods

2.1. Study Area: The Permian Basin

The Permian basin is the largest O&G production region in the United States and the second-largest hydrocarbon-bearing area in the world. It is located in the south of the country, between Texas and New Mexico with an extension of 160,000 km² (Robertson et al., 2020). The basin is formed by different production sub-basins that can be classified as the most productive sub-basins (Delaware, Central, and Midland) and the ones still in exploration and development (Ozona Arc and Valverde) (Figure 2).

The oil production in the basin increased from 900 thousand barrels per day in 2010 to 4,177 thousand barrels per day in 2019 (U.S. Energy Information Administration, 2019). In the case of gas, the increase in production over the last 10 years has been from 4,000 million cubic feet per day to 17,000 million cubic feet per day at the beginning of 2020.

O&G exploration in the Permian basin is primarily done using non-conventional techniques, including hydraulic fracturing (fracking) and horizontal drilling. The region is covered with thousands of active and abandoned wells, while continuously new wells are drilled. During drilling, heavy machinery, and engines, which amongst others emit NO_x , are used to create hydraulic fractures. Drilling is the step in the oil and gas production which can respond relatively quickly to market changes (Dix et al., 2020). A drop in demand therefore results in a strong decrease in the rig count (see Figure 1).

2.2. Data

This research is based on multiple sources of data: (a) NO₂ tropospheric vertical column densities, (b) wind data, (c) O&G production rates, and (d) drilling activity.





Figure 2. Monthly average of gas (a) and oil (b) production distribution in the Permian basin in 2020. The numbers are related to the different sub-basins of the Permian: 1-Midland, 2-Central, 3-Delaware, 4-Ozona Arc, and 5-Valverde.

We have used TROPOMI NO₂ Version 1.3 data (European Space Agency, 2021; Sentinel-5P Pre-Operations Data Hub, 2021; Van Geffen et al., 2020) with a quality indicator (qa_value) of at least 0.75. This ensures filtering for good quality NO₂ tropospheric columns for conditions with low cloud fractions. The NO₂ TROPOMI data processors were upgraded after October 2020, which causes jumps in the data records. Therefore, we limit our analysis to data until October 2020.

Monthly gas production, oil production, and drilling activity data were acquired from Enverus DrillingInfo (https://www.enverus.com), last accessed (30 October 2021).

In the case of drilling activity, the data provided by ENVERUS is reported in "drilling days." Drilling-days-counts can implicate more than one rig in the well pad, for example, three rigs drilling 20 days each, will be 60 days of drilling in total. The maximum number of rigs per well pad is 9, therefore the maximum number of days per month of drilling activity is 270.

2.3. Processing of TROPOMI Data

The TROPOMI orbit-based data files were filtered to the area of interest, in this case, the Permian basin, using the shapefile provided by the U.S. Energy Information Administration (E.I.A, 2017). From these files daily gridded NO_2 data was produced on $0.025^\circ \times 0.025^\circ$ (approximately 2.5 km \times 2.1 km) latitude/longitude grids. Subsequently, monthly mean and median NO_2 data were derived from the daily data, for the period January 2019 to October 2020. This study is focus in three different monthly time series: (a) COVID-19 lockdown (March–April), (b) whole year (January–October), and (c) before and after the lockdown (January–June).

2.4. Divergence Method for NO₂ Emissions Estimation

To estimate the NO₂ emissions using the total NO₂ columns, the divergence method was applied following the steps described by Beirle et al. (2019) and Dix et al. (2022). The divergence method was applied to the daily gridded NO₂ data set. For the wind speed and direction, the average value over the boundary layer height is used. Both the wind data and the boundary layer height are obtained by spatial and temporal interpolation in the publicly available ECMWF (European Centre for Medium Range Weather Forecasts) ERA 5 data set (ECMWF, 2021; Hersbach, H. et al., 2018). For the boundary layer height, a minimum value of 200 m was applied. The NO₂ lifetime is estimated from the average OH (hydroxyl radical) concentration in the boundary layer, using:

 $t_{NO2} = ([OH] k)^{-1}$



698996

2023, 13, Dov

where t_{NO2} is the lifetime in s, [OH] in mol cm⁻³ and k a constant of 3×10^{-11} s mol⁻¹ cm³ (Atkinson et al., 2004; IUPAC, 2022). The concentration of OH was obtained for the CAMS (Copernicus Atmosphere monitoring Service) global forecast (Peuch et al., 2022). Dix et al. (2022) concluded that using the divergence method, NO_x emissions from oil and gas production areas can be calculated on an annual basis within an uncertainty of 50%. Furthermore, the NO_x emission maps derived with the divergence method provide a high level of spatial detail with a precision that likely exceeds the accuracy of the total NO_x quantification.

2.5. Source Attribution and Spatial Analysis

A source attribution analysis was performed for the entire Permian basin and separately for each sub-basin. The method of this analysis is to combine the TROPOMI NO_2 emission data with the activity information. Each grid box in the monthly data set is classified using a combination of the NO_2 emission data and the activity data. To this end, 30 classes were defined (Figure S2, Tables S1 and S2 in Supporting Information S1) with the combination of different thresholds for production, drilling, and emission data. The segmentation of the different thresholds was selected having into account the average, the standard deviation, and the maximum for each data set. For example, for the emissions segmentation:

Mean emissions in the Permian (Mean) $\pm 2.3 \, e^{-9} \, mol \cdot s^{-1} \cdot m^2$

Standard deviation (SD) : $1.9 e^{-9} \text{ mol} \cdot \text{s}^{-1} \cdot \text{m}^2$

Maximum Emissions (MaxE) : 1.47 e^{-8} mol \cdot s⁻¹ \cdot m²

The split of the thresholds:

Average emissions : Mean – SD \leq Emissions \leq Mean + SD

Low emissions : Emissions \leq Mean – SD

High emissions : Mean + SD < Emissions < MaxE – [(Mean + SD) - (MaxE)/2]

Very high emissions : Emissions > MaxE - [(Mean + SD) - (MaxE)/2]

	Threshold (mol·s ^{-1} ·m ²)	Description
Emissions	Emissions $\leq 3.5 e^{-10}$	Low emissions
	$3.5 e^{-10} \le \text{Emissions} \le 4.3 e^{-9}$	Average emissions
	$4.3 e^{-9} < \text{Emissions} < 9.3 e^{-9}$	High emissions
	Emissions > 9.3 e^{-9}	Very high emissions

In addition, emissions located in cities (Texas Department of Transportation, 2021c; U.S. Geological Survey, 2019), airports (Analysis Center Earth Data, 2019; Texas Department of Transportation, 2021a), highways and, secondary roads (Program New Mexico 911 (NM911), 2021; Texas Department of Transportation, 2021b) were omitted for the reason that significant emissions sources other than from the O&G industry are expected for these locations. Grid boxes for which the emission data contained fill values were classified as "no production/drilling."

3. Results

3.1. Spatial Distribution of NO₂ Tropospheric Concentrations During the COVID-19 Lockdown in the Permian Basin

The monthly median of NO_2 is shown in Figure 3 for the months of January to June for 2019 and 2020. NO_2 concentrations are higher in winter due to the longer lifetime of NO_2 during these months. The impact of COVID-19 on NO_2 concentrations is analyzed by comparing the same periods between 2019 and 2020. The period

4 of 12



10.1029/2023JD038566



Figure 3. Time series of median nitrogen dioxide tropospheric column from January to June 2019 and 2020 in the Permian basin.

January to March 2020 is not impacted by any COVID-19 policy measures, whereas during the period April to June 2020, there were policy measures in the region as well as around the world. In this analysis, we used a 3-month period to reduce the effects of differences in meteorology in 2019 versus 2020.

As compared to the same period in 2019, the NO₂ tropospheric concentrations show an increase during the winter months of 2020 (from January 2020 to March 2020) of an average 9%, where Midland and Central sub-basins show the largest increases (13.3% and 13.2%, respectively). A decrease is found during the lockdown months (April to June 2020) of an average of -4% (Table 1; Figure S3 in Supporting Information S1), which is quite similar across the most productive sub-basins (Delaware, Central, and Midland).

Sources of the variability of NO_2 concentrations include meteorology (cloud cover, wind, etc.), the atmospheric chemistry (chemical reaction rates and photolysis) which can change the lifetime of NO_2 and measurement noise

Table 1

Changes in NO_2 Concentration Average, Minimum, Maximum, Interquartile Range, Average $\pm 95\%$ Confidence Interval, and Standard Deviation Across All the Sub-Basins in the Permian Basin

Sub-basin	Average % of change (%)	Min % of change (%)	Max % of change (%)	Interquartile range (50%) (%)	Average and 95% confidence interval $(NO_2 \text{ in } \mu \text{mol}/\text{m}^2)$	Standard deviation of the mean (NO ₂ in μ mol/m ²)
Delaware	-4.4	-5	0	-5.9	$5.1 \le 18.1 \le 33.0$	7.0
Midland	-4.6	21	-28	-0.3	$5.1 \leq 18.6 \leq 33.0$	7.6
Central	-4.6	-76	12	-3.3	$4.1 \leq 18.0 \leq 32.7$	7.1
Ozona Arc	-6.3	30	0	8.4	$5.1 \leq 14.6 \leq 33.0$	5.9
Valverde	0.0	0	55	1.3	$1.3 \leq 14.1 \leq 26.7$	7.5

Note. The percentages are related to April to June 2020 with respect the April to June 2019.



10.1029/2023JD038566



Figure 4. Median nitrogen dioxide emissions in 2019 (a) and 2020 (b) from January to October obtained using TROPOMI NO₂ tropospheric columns and the divergence method.

from the instrument. The similarity in the standard deviations of all the sub-basins during the lockdown period demonstrates that the data variability is comparable. The uncertainty in the measurement of the mean in all the basins is comparable with only the Delaware basin having a 95% confidence interval higher than the other basins (Table 1).

3.2. Analysis of NO_2 Emissions During the COVID-19 at Basin and Sub-Basin Level in the Permian Basin

The median of NO_2 emissions derived from the TROPOMI concentrations using the divergence method, are shown in Figure 4 for the period January to October 2019 and January to October 2020. The regions with the highest NO_2 emissions are the ones that show the major reductions in 2020 with respect to the same period in 2019.

The variability of the emissions between February and June 2019 (Figure S8 in Supporting Information S1; Figure 5) is considerably smaller as compared to the concentrations for the same period (Figure 3). This indicates that the chemical loss of NO_2 , which is the main driver for the seasonal variations in the concentrations, and the meteorology are generally well represented in the divergence method. The emissions for 2020 show a different behavior, with a strong decrease after March 2020. Comparing NO_2 median emissions during the COVID-19 lockdown period (April to June 2020) with the same period in 2019, an average of 30% decrease was observed in all the sub-basins that compose the Permian basin, with Midland as the sub-basin with the largest decrease (-33%) (Figure S8 in Supporting Information S1).

Reference regions outside the boundaries of the Permian basin with background conditions (areas without the presence of O&G industrial activities) were also analyzed (Figure S4 in Supporting Information S1). The NO_2 median emissions decreased between 1% and 10%, except for regions on the East of the Permian basin where large cities like Dallas, San Antonio, or Austin are located. This analysis of the background regions verifies that the decrease of the NO_2 emissions is not related to artifacts in the data or the methods, but to reduced emissions in the area (an average decrease of 30%). The decreased emissions during the lockdown period were found in regions dominated by O&G industrial activities, as well as in areas with emissions not dominated by the O&G, such as the Interstate 20 Frontage Road and the major cities of the basin (Pecos, Odesa, Midland, and Carlsbad).





Figure 5. Histograms of the NO₂ emissions for the most productive sub-basins (a, b, and c) and the less productive (d and e) in the Permian basin during the period April-May in 2019 and 2020. The percentages in the plot title are related to the reduction of NO₂ emissions between 2019 and 2020 (SD: Standard deviation).

3.3. Source Attribution of NO_2 Emissions in the Permian Basin Before and During the COVID-19 Lockdown

 NO_2 emissions calculated with the divergence method were combined with O&G production, drilling, and other activities. Figure 6 shows the mean emission for the grid boxes associated with these activity categories, as a function of the month in 2019 and 2020. NO_2 associated with production and drilling in the same place, and places with only drilling activities were the most impacted during and after the COVID-19 lockdown. Decreases of 67% with respect to 2019 are found during the period from April to June 2020 (Table 2). According to Dix et al. (2020), the number of active drilling wells responds quickly to economic changes but places with only production activity more gradual, which can explain the large decrease in places with significant drilling activities.

In a longer time-series analysis (January to October 2020), the impact of COVID-19 was also most pronounced in production and drilling regions (-32%), and only an impact of -3% decrease for the other categories (Table 2), mainly related to the rapid response of the industry (reduction of drilling activities and production intensity). The entire Permian basin showed an impact of -7% if we compare 2020 with 2019, but if the comparison is limited to the lockdown period changes in emissions of up to -19% was found. Excluding the impact of other sources of emissions, the effect of the COVID-19 lockdown was -24% for NO₂ emissions which were associated with the O&G industry.

The same analysis was performed for the main sub-basins of the Permian basin (Table S3 in Supporting Information S1). For the period January to October, NO_2 emissions associated to places with production and drilling



10.1029/2023JD038566



Figure 6. Normalized NO_2 emission for each month in areas with only production, production and drilling, drilling, and other activities not related to the O&G. NO_2 emissions were normalized by the number of pixels attributed to each activity on the entire Permian basin.

activity on the same site showed an important decrease in the Delaware, Midland, and Central sub-basins. As the number of locations with NO_2 emissions associated to only production activities did not change significantly (Figures S5 and S9 in Supporting Information S1), this decrease appears to be driven by the intensity of the activities.

Finally, we analyzed the production rates in locations with high levels of NO₂ emissions (4.3 e⁻⁹ mol·s⁻¹·m⁻² < NO₂ emissions < 9.3 e⁻⁹ mol·s⁻¹·m⁻² of NO₂) and places with average levels (3.5 e⁻¹⁰ mol·s⁻¹·m⁻² < emissions < 4.3 e⁻⁹ mol·s⁻¹·m⁻²) during the COVID-19 lockdowns (Figure 7; Figures S7 and S9 in Supporting Information S1). As can be seen in Figure 7 and in Figure S9 in Supporting Information S1, in March 2020 average NO₂ emissions from oil and gas production locations were mainly associated with low production places (Production < 5,000 Mcf/month in the case of gas and between 0 < Production < 800 barrels a month for oil) and high NO₂ emissions to locations with high production (40,000 Mcf/month < Production < 600,000 Mcf/month for gas and 12,000 < Production < 100,000 barrels in the case of oil). With the COVID-19 lockdown, a significant decrease in high NO₂ emissions (4.3 e⁻⁹ mol·s⁻¹·m⁻² < NO₂ emissions < 9.3 e⁻⁹ mol·s⁻¹·m⁻² of NO₂) associated with places with high production rates can be observed in the Delaware basin and Central basin, and an increase in average NO₂ emissions (3.5 e⁻¹⁰ mol·s⁻¹·m⁻² ≤ emissions ≤ 4.3 e⁻⁹ mol·s⁻¹·m⁻²) in the same locations. In May 2020, just after the peak of the lockdown, the total NO₂ emission from places classified as high emitter location were reduced from 0.17 mol·s⁻¹ of NO₂ to 0.03 mol·s⁻¹ in the Permian basin (Figure 7; Figure S7 in Supporting

Table 2

Total NO₂ Emission Rate of Each O&G Activity in the Permian Basin in 2019 and 2020 (January to October and April to June) for Areas With Only Production, Production + Drilling, Drilling and Other Activities

		Production	Production + drilling	Drilling	Others	Total Permian
January to October	Permian basin 2019 (mol·s ⁻¹ of NO ₂)	2.17	0.42	0.009	1.47	4.08
	Permian basin 2020 (mol·s ⁻¹ of NO ₂)	2.09	0.27	0.009	1.42	3.81
	% Difference	-3.7%	-32%	-2.8%	-3.4%	-7%
April to May	Permian basin 2019 (mol·s ⁻¹ of NO ₂)	0.70	0.135	0.003	0.48	1.33
	Permian basin 2020 (mol·s ⁻¹ of NO ₂)	0.60	0.044	0.001	0.43	1.07
	Decrease due to lockdown (mol·s ⁻¹ of NO ₂)	0.11	0.095	0.002	0.06	0.257
	% Difference	-16%	-67%	-66%	-12%	-19%

Note. The emissions were normalized to the number of pixels for each category.





Figure 7. NO₂ high emissions (4.3 e⁻⁹ mol·s⁻¹·m⁻² < NO₂ emissions < 9.3e⁻⁹ mol·s⁻¹·m⁻²) (Bottom) and average emissions (3.5 e⁻¹⁰ mol·s⁻¹·m⁻² ≤ emissions ≤ 4.3 e⁻⁹ mol·s⁻¹·m⁻²) (Top) related to gas production rates in the Permian basin from March to May 2020.

Information S1). Although gas and oil production were analyzed separately the results related to the emissions attribution are similar because in the Permian basin most of the wells can produce oil and gas at the same time.

4. Discussion

The observed decrease NO_2 concentrations and emissions in the Permian basin corresponded in time with the decline in demand for fossil fuels, due to the COVID-19 lockdowns and travel bans. For the period between April and June 2020, we report decreases of NO_2 tropospheric column concentrations of 4% and emissions derived using the divergence method of 30%, compared to the same period in 2019. This difference in reduction between concentrations and emissions in the same location has been also reported by (Misra et al., 2021). The average decrease (4%) in tropospheric concentrations from March to May 2019 was also observed by Qu et al. (2021), justifying this weaker decrease in NO_2 concentrations because of the inability of TROPOMI to capture the decrease in seasonal progression periods due to the dampening effect from the NO_2 background. This research



also demonstrates the ability of TROPOMI to capture the larger reductions of NO_2 in small areas where there are high levels of NO_2 emissions (as it was observed in the maximum values in the Midland basin—28%) which is also in agreement with Qu et al. (2021). Although other uncertainties can be present in concentrations (e.g., measurement noise, chemical reaction rates, or meteorology), the use of temporal and spatial averaging during only 3 months of the COVID-19 lockdown can reduce the random errors and impacts of the seasonal variations of the NO_2 lifetime.

The decreases in emissions are driven by decreased O&G industry activity which we quantified as a decrease of 68% in drilling activities and 16% in production activities. However, not all O&G activities decreased in the same way: in the case of production, the number of active wells remained the same (Table S4 in Supporting Information S1), and just reducing the production rate (-13% respect to 2019) caused a 16% reduction of NO₂, and consequently a reduction in the emission rate per meter squared (from high emission rates (4.3 e⁻⁹ mol·s⁻¹·m² < NO₂ emissions < 9.3 e⁻⁹ mol·s⁻¹·m² of NO₂) to what it was categorized as average emission rates (3.5 e⁻¹⁰ mol·s⁻¹ m² ≤ emissions ≤ 4.3 e⁻⁹ mol·s⁻¹·m²). The relationship between changes in production rates and NO₂ enhancements detected by OMI (Ozone Monitoring Instrument) was demonstrated by Majid et al. (2017), indicating an increase of approximately 0.15 e⁻¹⁵ mol cm⁻² of NO₂ for an increase of 1 million barrels daily.

Drilling activities, especially the ones located at the same location as production, reduced the number of active drilling rigs drastically instead of decreasing the drilling intensity (reducing the number of days drilling per month). The absolute largest reduction in NO₂ emissions was associated with locations with only production $(0.108 \text{ mol} \cdot \text{s}^{-1} \text{ of NO}_2)$. However, the number of emitting locations (48% (Figure S6 in Supporting Information S1)) is much higher than the places having drilling + production on the same location (7% (Figure S6 in Supporting Information S1)). In the case of drilling activities in the same place as production, the reduction of NO₂ emissions (0.091 mol·s⁻¹) was similar to the one associated with places with only production activity, but for 40% fewer locations (Figure S6 in Supporting Information S1). Therefore, the fact that having a similar level of emissions but much less locations, makes the reduction of drilling activity a powerful mitigation action to reduce NO₂ emissions in the Permian basin.

Several authors have shown the relationship between lockdown policies and the decrease in NO₂ at different scales of analysis: at the global scale (Venter et al., 2020; Zhang et al., 2021), country (Archer et al., 2020; Qu et al., 2021) and cities (Barré et al., 2020). In the case of urban areas, Barré et al. (2020) shows an overall reduction in European cities between 20% and 40%. Misra et al. (2021) presented the decrease in NO₂ emissions in urban areas (-73%) and power plants (-53%) in the North of India with the use of the divergence method. The divergence method has been also applied to detect missing NO₂ emissions from the O&G industry inventories (Dix et al., 2021), demonstrating the suitability of the method to detect emissions caused by the O&G industry. The impact of the COVID-19 lockdowns in the Permian basin, which is predominantly driven by changes in O&G industrial activities, is therefore on the low side as compared to reductions in urban areas.

5. Conclusions

The exceptional scenario of the COVID-19 pandemic created an opportunity to assess the contribution of the oil and gas sector to nitrogen dioxide emissions in the biggest production area of the United States, the Permian basin. This research shows for the first time the impact of reducing the activity source of NO_2 emissions in a dynamic spatial and time analysis of an area characterized by widespread source points and illustrating a different scenario from studies in urban areas (Barré et al., 2020; Zhang et al., 2021).

The influence of lockdown policies had an impact on global energy trends, causing a major drop in oil and gas demand. This decrease in the oil and gas activity also observed in the Permian basin (13% of production and 68% in drilling activities) caused a significant reduction in NO₂ emissions (~30%) and a weaker decrease in tropospheric concentrations (-4%) masked by the impact of the NO₂ background. However, stronger reductions of NO₂ tropospheric concentrations can be observed in smaller areas where the emissions suffered a strong decrease. During the lockdown period, locations with both production and drilling activity at the same place showed a similar decrease in NO₂ emissions as places with only production, even having less amount of source points across the Permian basin. The reduction of drilling activity can be a good strategy for NO₂ emission reduction in the Permian basin.

21698996,



21698996,

2023, 13, Downlo

ided from https:/

This research has revealed the potential of TROPOMI NO_2 observations and derived emissions to track variations of the O&G production and drilling activities.

Data Availability Statement

Main data sets used in this publication are:

- TROPOMI NO₂ Version 1.3 data available at https://s5phub.copernicus.eu/dhus/#/home.
- ERA-5 meteorological information available at https://www.ecmwf.int/en/forecasts/datasets/ reanalysis-datasets/era5.

References

Analysis Center Earth Data. (2019). New Mexico public airports.

- Archer, C. L., Cervone, G., Golbazi, M., Al Fahel, N., & Hultquist, C. (2020). Changes in air quality and human mobility in the USA during the COVID-19 pandemic. *Bulletin of Atmospheric Science and Technology*, 1(3–4), 491–514. https://doi.org/10.1007/s42865-020-00019-0
- Atkinson, R., Baulch, D. L., Cox, R. A., Crowley, J. N., Hampson, R. F., Hynes, R. G., et al. (2004). Evaluated kinetic and photochemical data for atmospheric chemistry: Volume I—Gas phase reactions of O_x, HO_x, NO_x and SO_x species. *Atmospheric Chemistry and Physics*, 4(6), 1461–1738. https://doi.org/10.5194/acp-4-1461-2004
- Barré, J., Petetin, H., Colette, A., Guevara, M., Peuch, V.-H., Rouil, L., et al. (2020). Estimating lockdown induced European NO₂ changes. Atmospheric Chemistry and Physics Discussions (2), 1–28. https://doi.org/10.5194/acp-2020-995
- Beirle, S., Borger, C., Dörner, S., Eskes, H., Kumar, V., De Laat, A., & Wagner, T. (2019). Catalog of NO_x emissions from point sources as derived from the divergence of the NO₂ flux for TROPOMI (no. 2, pp. 1–28).
- Ding, J., van der A, R. J., Eskes, H. J., Mijling, B., Stavrakou, T., van Geffen, J. H. G. M., & Veefkind, J. P. (2020). NO_x emissions reduction and rebound in China due to the COVID-19 crisis. *Geophysical Research Letters*, 47(19). https://doi.org/10.1029/2020GL089912

Dix, B., de Bruin, J., Roosenbrand, E., Vlemmix, T., Francoeur, C., Gorchov-Negron, A., et al. (2020). Nitrogen oxide emissions from U.S. oil and gas production: Recent trends and source attribution. *Geophysical Research Letters*, 47(1). https://doi.org/10.1029/2019GL085866

- Dix, B., Francoeur, C., Li, M., Serrano-Calvo, R., Levelt, P. F., Veefkind, J. P., et al. (2022). Quantifying NO_x emissions from U.S. oil and gas production regions using TROPOMI NO₂. ACS Earth and Space Chemistry, 6(2), 403–414. https://doi.org/10.1021/acsearthspacechem.1c00387
- Dix, B., Francoeur, C., Mcdonald, B., Serrano, R., Veefkind, P., & De Gouw, J. (2021). NO_x emissions from U.S. oil and gas production using TROPOMI NO₂ and the divergence method (2), 13554.

ECMWF. (2021). ERA-5 [Dataset]. ECMWF. Retrieved from https://www.ecmwf.int/en/forecasts/datasets/reanalysis-datasets/rea5

- E.I.A. (2017). PermianBasin_Extent_201712.shp. Energy Information Administration. Retrieved from https://www.eia.gov/maps/maps.htm European Space Agency. (2021). Copernicus Sentinel-5P (processed by ESA), 2021, TROPOMI level 2 nitrogen dioxide total column products. Version 02. https://doi.org/10.5270/S5P-9bnp8q8
- Gaubert, B., Bouarar, I., Doumbia, T., Liu, Y., Stavrakou, T., Deroubaix, A., et al. (2021). Global changes in secondary atmospheric pollutants during the 2020 COVID-19 pandemic. *Journal of Geophysical Research: Atmospheres*, *126*(8), 1–22. https://doi.org/10.1029/2020JD034213
- Hersbach, H., Bell, B., Berrisford, P., Biavati, G., Horányi, A., Muñoz Sabater, J., et al. (2018). ERA5 hourly data on single levels from 1979 to present. Copernicus Climate Change Service (C3S) Climate Data Store (CDS). https://doi.org/10.24381/cds.adbb2d47
- Honour, S. L., Bell, J. N. B., Ashenden, T. W., Cape, J. N., & Power, S. A. (2009). Responses of herbaceous plants to urban air pollution: Effects on growth, phenology and leaf surface characteristics. *Environmental Pollution*, 157(4), 1279–1286. https://doi.org/10.1016/j.envpol.2008.11.049
- Huang, S., Li, H., Wang, M., Qian, Y., Steenland, K., Caudle, W. M., et al. (2021). Long-term exposure to nitrogen dioxide and mortality: A systematic review and meta-analysis. *Science of the Total Environment*, 776, 145968. https://doi.org/10.1016/j.scitotenv.2021.145968
- IUPAC. (2022). IUPAC task group on atmospheric chemical kinetic data evaluation. Retrieved from https://iupac.aeris-data.fr/en/home/ Keller, C. A., Evans, M. J., Emma Knowland, K., Hasenkopf, C. A., Modekurty, S., Lucchesi, R. A., et al. (2021). Global impact of COVID-19 restrictions on the surface concentrations of nitrogen dioxide and ozone. *Atmospheric Chemistry and Physics*, 21(5), 3555–3592. https://doi. org/10.5194/acp-21-3555-2021
- Le Quéré, C., Jackson, R. B., Jones, M. W., Smith, A. J. P., Abernethy, S., Andrew, R. M., et al. (2020). Temporary reduction in daily global CO₂
- emissions during the COVID-19 forced confinement. *Nature Climate Change*, 10(7), 647–653. https://doi.org/10.1038/s41558-020-0797-x Majid, A., Martin, M. V., Lamsal, L. N., & Duncan, B. N. (2017). A decade of changes in nitrogen oxides over regions of oil and natural gas
 - activity in the United States. *Elementa*, 5. https://doi.org/10.1525/elementa.259
- Misra, P., Takigawa, M., Khatri, P., Dhaka, S. K., Dimri, A. P., Yamaji, K., et al. (2021). Nitrogen oxides concentration and emission change detection during COVID-19 restrictions in North India. *Scientific Reports*, 11(1), 1–11. https://doi.org/10.1038/s41598-021-87673-2
- Peuch, V. H., Engelen, R., Rixen, M., Dee, D., Flemming, J., Suttie, M., et al. (2022). The Copernicus atmosphere monitoring service from research to operations. *Bulletin of the American Meteorological Society*, *103*(12), E2650–E2668. https://doi.org/10.1175/BAMS-D-21-0314.1 Program New Mexico 911 (NM911). (2021). New Mexico roads centerlines. Retrieved from http://rgis.unm.edu/rgis6/
- Qu, Z., Jacob, D. J., Silvern, R. F., Shah, V., Campbell, P. C., Valin, L. C., & Murray, L. T. (2021). US COVID-19 shutdown demonstrates importance of background NO₂ in inferring NO_x emissions from satellite NO₂ observations. *Geophysical Research Letters*, 48(10), 1–8. https://doi. org/10.1029/2021GL092783
- Robertson, A. M., Edie, R., Field, R. A., Lyon, D., McVay, R., Omara, M., et al. (2020). New Mexico Permian basin measured well pad methane emissions are a factor of 5–9 times higher than U.S. EPA estimates. *Environmental Science and Technology*, 54(21), 13926–13934. https:// doi.org/10.1021/acs.est.0c02927
- Rume, T., & Islam, S. M. D. U. (2020). Environmental effects of COVID-19 pandemic and potential strategies of sustainability. *Heliyon*, 6(9), e04965. https://doi.org/10.1016/j.heliyon.2020.e04965
- Sentinel-5P Pre-Operations Data Hub. (2021). TROPOMI NO₂ version 1.3 [Dataset]. Copernicus EU. Retrieved from https://s5phub.copernicus. eu/dhus/#/home
- Tang, R., Huang, X., Zhou, D., Wang, H., Xu, J., & Ding, A. (2021). Global air quality change during the COVID-19 pandemic: Regionally different ozone pollution responses COVID-19. Atmospheric and Oceanic Science Letters, 14(4), 100015. https://doi.org/10.1016/j.aosl.2020.100015

Acknowledgments

We acknowledge the free use of the tropospheric nitrogen dioxide column data available on the Copernicus S5p Open Access Hub: https://scihub. copernicus.eu/. We acknowledge helpful discussions with Rutger IJzermans, Richard Eyers, Bas van de Kerkhof, David Randell, and Matthew Jones. This work was financially supported by Shell Global Solutions International B.V. with grant reference number CW284766. Drilling and production activity data were provided by Enverus DrillingInfo. B.D. and J.dG were financially supported by the NASA ACMAP program under award number 80NSSC19K0979, and by the Rocky Mountain Institute and Blue Sky Resources

11 of 12

Texas Department of Transportation. (2021a). *Texas airports*. Texas Department of Transportation. Retrieved from https://gis-txdot.opendata. arcgis.com/datasets/texas-airports/explore?location=31.173825%2C-100.059833%2C6.70

Texas Department of Transportation. (2021b). TxDOT roadways.

- Texas Department of Transportation. (2021c). TxDOT urbanized areas. Retrieved from https://gis-txdot.opendata.arcgis.com/datasets/ txdot-urbanized-areas/explore?location=31.134058%2C-100.180538%2C6.70
- Turner, A. J., Kim, J., Fitzmaurice, H., Newman, C., Worthington, K., Chan, K., et al. (2020). Observed impacts of COVID-19 on urban CO2 emissions. *Geophysical Research Letters*, 47(22), 1–6. https://doi.org/10.1029/2020GL090037
- U S Energy Information Administration. (2019). International energy outlook 2019. IEA. https://doi.org/10.5860/choice.44-3624

U.S. Geological Survey. (2019). Populated places 2016.
Van Geffen, J., Folkert Boersma, K., Eskes, H., Sneep, M., Ter Linden, M., Zara, M., & Pepijn Veefkind, J. (2020). S5P TROPOMI NO₂ slant column retrieval: Method, stability, uncertainties and comparisons with OMI. *Atmospheric Measurement Techniques*, 13(3), 1315–1335. https://doi.org/10.5194/amt-13-1315-2020

- Veerkind, J. P., Aben, I., McMullan, K., Förster, H., de Vries, J., Otter, G., et al. (2012). TROPOMI on the ESA Sentinel-5 Precursor: A GMES mission for global observations of the atmospheric composition for climate, air quality and ozone layer applications. *Remote Sensing of Environment*, 120, 70–83. https://doi.org/10.1016/j.rse.2011.09.027
- Venter, Z. S., Aunan, K., Chowdhury, S., & Lelieveld, J. (2020). COVID-19 lockdowns cause global air pollution declines. Proceedings of the National Academy of Sciences of the United States of America, 117(32), 18984–18990. https://doi.org/10.1073/pnas.2006853117
- Zhang, H., Lin, Y., Wei, S., Loo, B. P. Y., Lai, P. C., Lam, Y. F., et al. (2021). Global association between satellite-derived nitrogen dioxide (NO₂) and lockdown policies under the COVID-19 pandemic. *Science of the Total Environment*, 761(2), 144148. https://doi.org/10.1016/j. scitotenv.2020.144148